Author's response to reviews

Title: Oxycodone-induced cholestatic hepatitis: a case report

Authors:

   Vincent Ho (vincent_ho@health.qld.gov.au)
   Maxwell Stewart (maxwell_stewart@health.qld.gov.au)
   Peter Boyd (peter_boyd@health.qld.gov.au)

Version: 3 Date: 8 January 2008

Author's response to reviews: see over
Dear Editor,

Thank you for the constructive comments of the reviewers.

In response to the reviewers:

1) Formal recognition of the point that our patient had not been taking any traditional medicines is well-made and we have amended a sentence in the case report to read ‘He had not taken any other medications since discharge including over-the-counter medications, herbal or traditional medicines.’

2) Our patient has never consumed alcohol and this is clearly stated in the sentence ‘He had never consumed alcohol and there was no recent travel history.’ We have thus left this unchanged.
